Author’s response to reviews

Title: Routine care provided by specialists to children and adolescents in the United States: 2002-2006

Authors:

Jose M Valderas (jose.valderas@manchester.ac.uk)
Barbara Starfield (bstarfie@jhsphs.edu)
Christopher B Forrest (FORRESTC@email.chop.edu)
Luis Rajmil (lrajmil@imim.es)
Martin Roland (martin.roland@medschl.cam.ac.uk)
Bonnie Sibbald (bonnie.sibbald@manchester.ac.uk)

Version: 4 Date: 13 November 2009

Author’s response to reviews:

Dear Editor,

Please find below details of the formatting changes introduced to the manuscript:

Formatting changes to be made:

----------------------------------------
Major revisions (we require the author to make these changes)
References: References should be cited in the text using consecutive numbers in square brackets. In addition, reference 34 is not currently cited.

DONE

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Job titles: Please remove these from the title page.

DONE

Affiliations: Please remember to include the country in the affiliation details.

DONE

Abbreviations: Please format these in to a paragraph.

DONE

Placement text: Please remove all table placement text.

DONE
References: the reference list should contain up to 30 authors' names for each citation. The term 'et al.' should not be used.

DONE

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

DONE